Author's response to reviews

Title: N-acetyltransferase 8, a positional candidate for blood pressure and renal regulation: resequencing, association and in silico study

Authors:

Peeter Juhanson (peeter.juhanson@ut.ee)
Katrin Kepp (katike@ut.ee)
Elin Org (elin@ut.ee)
Gudrun Veldre (gveldre@ut.ee)
Piret Kelgo (piret.kelgo@hot.ee)
Mai Rosenberg (mai.rosenberg@kliinikum.ee)
Margus Viigimaa (margus.viigimaa@regionaalhaigla.ee)
Maris Laan (maris@ebc.ee)

Version: 4 Date: 9 April 2008

Author's response to reviews: see over
To: The BioMed Central Editorial Production Team  
BMC-series journals, *BMC Medical Genetics*

Response to requested formatting changes

Re: MS# 1499780885165491
Juhanson et al “N-acetyltransferase 8, a positional candidate for blood pressure and renal regulation: resequencing, association and *in silico* study”

Tartu, Estonia
09.04.2008

Dear *BMC Medical Genetics* Editors,

Dear The BioMed Central Editorial Production Team,

We have performed the requested formatting of the References (references to websites addresses and to multi-author studies) to correspond the style of *BMC* journals.

Yours sincerely,

Maris Laan,

Corresponding author

Research professor

Institute of Molecular and Cell Biology, University of Tartu